\input epsf
\input graphicx

\magnification 1400
\parindent 0pt
\parskip 10pt
\pageno=1
\nopagenumbers
\hoffset -0.4truein
\voffset -0.4truein
\hsize 7truein
\vsize 10.7truein
\def\u{\vskip -15pt}
\def\v{\vfill}


Quiz 2:
%%\vskip -20pt

1.) Write as a single simplified logarithm:  
$$ln(1)ln(3/5) + 2ln(3)- ln(e)ln(4) = \underline{\hskip 1.3in}$$
\v
2.) Graph the following

1a)  $y = e^x$ \hfil (b) $y = e^{-x}$ \hfil (c) $y = ln(x)$

\includegraphics[width=18ex]{graph}
 \hfil 
 \includegraphics[width=18ex]{graph}
 \hfil 
 \includegraphics[width=18ex]{graph}
 
3.)  Give an example of an initial value problem with multiple solutions.
\v
4.)  Define:  A function $f$ is linear if
\v


\end 

\end
